Author's response to reviews

Title: Pseudo tumoral tracheobronchial amyloidosis mimicking asthma: A case report

Authors:

Mounia Serraj (moonserraj@yahoo.fr)
Imane Kamaouï (kamaouiiimane@yahoo.fr)
kawtar Znati (kawtarznati@yahoo.fr)
salma Kouara (kouarasalma@gmail.com)
ferdaous sahnoune (ferdaous.05@hotmail.com)
bouchra amara (amarabouchra@gmail.com)
mohammed el biaze (melbiaze@gmail.com)
siham tizniti (sihamtizniti@yahoo.fr)
afaf amarti (amartiafaf@yahoo.fr)
mohammed chakib benjelloun (mcbenjelloun@yahoo.fr)

Version: 7 Date: 5 December 2011

Author's response to reviews: see over
Following comments, the manuscript was reviewed and language revised. In fact, I’ve submitted the manuscript to English native speaker.

Changes required to comply with the instructions for authors were also applied.

The study design was included in the title and the title was changed.

More details of medical history was also added.

Ethnicity of the patient was included in the abstract.

A literature review was performed and 2 tables was added to the text.